Author’s response to reviews

Title: An e-health driven laboratory information system to support HIV treatment in Peru: E-quity for laboratory personnel, health providers and people living with HIV/AIDS

Authors:

Patricia J García (pattyjannet@gmail.com)
Javier Vargas H (jvargas@ins.gob.pe)
Patricia Caballero N (pcaballe@ins.gob.pe)
Javier Calle V (callejavier@hotmail.com)
Angela M Bayer (angelabayerx@gmail.com)

Version: 4 Date: 26 November 2009

Author’s response to reviews:

We have made all of the formatting changes requested by the BioMed Central Editorial Production Team in an email dated November 17. All of the changes are included in the revised manuscript document.